Author's response to reviews

Title: Recruitment of young women to a trial of chlamydia screening - as easy as it sounds?

Authors:

Helen Atherton (helen.atherton07@imperial.ac.uk)
Debbie Banks (deborahbanks@gmail.com)
Ruth Harbit (ruthharbit@cantab.net)
Linzie Long (m0500727@sgul.ac.uk)
Fiona Chadd (m0500446@sgul.ac.uk)
Phillip Hay (phay@sgul.ac.uk)
Sally Kerry (sgjdgp@sgul.ac.uk)
Ian Simms (ian.simms@hpa.org.uk)
Pippa Oakeshott (oakeshot@sgul.ac.uk)

Version: 4 Date: 20 November 2007

Author's response to reviews: see over
Dear Editor

Thank you very much for agreeing to publish our manuscript. Here we detail our formatting changes as requested.

1) We have added a title page with the author list on one line, affiliations written in full and linked to each author via superscript numbers, corresponding author and a list of email addresses.

2) Keywords have been removed.

3) Author contributions have been placed towards the end of the manuscript preceding the acknowledgements.

4) The figure legend has been placed after the references.

5) An additional files section has been added at the end of the manuscript in the format recommended.

We very much hope these changes are satisfactory.

With Best Wishes

Yours sincerely

Helen Atherton